# **Two Types of Parasitic Assimilation**\*

Peter Jurgec

Meertens Institute, Amsterdam & Leiden University

### Abstract

This paper shows that consonant harmony and parasitic vowel harmony are more similar than previously assumed. I provide a unified and restrictive analysis of parasitic assimilation using feature spreading constraints. In particular, I attribute the differences between the attested and unattested patterns to two types of markedness constraints—alignment and agreement.

# 1. Introduction

The phonological literature standardly recognizes two complementing forces in assimilation. First, assimilation affects segments in specific positions. For example, vowel harmony targets syllable nuclei (vowels), umlaut and metaphony often target stressed syllables, whereas voicing assimilation targets codas. Second, assimilation involves similar segments. For instance, parasitic vowel harmony applies only if the trigger and the target display an additional similarity not shared by all vowels; consonant harmony applies to a subset of consonants that are in some respect like the trigger; voicing assimilation in some languages is observed as long as the trigger and target are either both stops or both fricatives.

These two forces limit the extent of the attested cross-linguistic variation. Their effects in parasitic vowel harmony and consonant harmony are the focus of this paper. The empirical insight is that the two patterns are more alike than previously assumed. The proposed analysis mirrors this similarity: all kinds of assimilation involve feature spreading governed by constraints in Optimality Theory (OT). The differences between the two patterns stem from an asymmetry between vocalic and consonantal targets.

This paper is organized as follows. Section 2 offers a short overview of two parasitic patterns: parasitic vowel harmony in Yowlumne and retroflex (consonant) harmony in Kalasha. Section 3 looks at the properties of non-parasitic assimilation. The main point is to show that targets of assimilation display an asymmetry: some positions are generally preferred compared to others. This effect is attributed to alignment constraints. Section 4 looks at parasitic vowel harmony in Yowlumne, whereas section 5 provides an analysis of Kalasha retroflex harmony. Both patterns are analyzed as spreading resulting from the interaction of alignment and agreement constraints. Section 6 compares the current analysis with one alternative. Section 7 concludes the paper.

The contribution of this paper is two-fold. First, I show that parasitic vowel harmony and consonant harmony are two variations of the same pattern—parasitic assimilation. Second, I provide a unified analysis of both phenomena based on feature spreading and two classes of OT constraints. This approach predicts parasitism in vowel and consonant harmony, but limits non-parasitic assimilation to vowel harmony.

# 2. Parasitic assimilation

Assimilation targets a natural class of segments. Vowel harmony, for example, typically targets all vowels in a word or some other prosodic domain. Vowels may act as targets even if the trigger is a consonant. What this suggests is that the phonological properties of the trigger most commonly play little role, what matters are the targets (Nevins 2010). Once we consider the phonological properties of the trigger, however, we get several additional patterns. In some cases, only a subset of segments with a particular underlying feature

<sup>\*</sup>For their comments and suggestions, I am indebted to Laura Downing, Martin Krämer, Bruce Morén-Duolljá, Curt Rice, Marc van Oostendorp, and two anonymous reviewers. I have also received valuable feedback from Akinbiyi Akinlabi, Paul de Lacy, Edward Flemming, Paula Houghton, Pavel Iosad, Kristen de Joseph, Kathrin Linke, Claire Moore-Cantwell, Alan Prince, Keren Rice, Franziska Scholz, Donca Steriade, Marijn van t' Veer, and Rachel Walker.

<sup>© 2013</sup> Peter Jurgec. *Nordlyd* 40.1: 108-135, special issue 'A Festschrift on the Occasion of X Years of CASTL Phonology and Curt Rice's L<sup>th</sup> Birthday', ed. by Sylvia Blaho, Martin Krämer and Bruce Morén-Duolljá. University of Tromsø. http://www.ub.uit.no/baser/nordlyd

triggers assimilation (McCarthy 2009, Mullin 2011, Jurgec 2011b). In other cases, assimilation depends on an additional similarity between a trigger and a target. This is the case in parasitic assimilation.

To illustrate, let us look at a classic example of parasitic vowel harmony (Steriade 1981, Archangeli 1985, Cole 1987, Cole and Trigo 1988, Kaun 1995; 2004, Krämer 2003) found in the Yowlumne dialect of Yokuts (formerly known as Yawelmani). Yowlumne has rounding harmony. Often, rounding harmony affects all vowels indiscriminately, as in Kyrgyz (Wurm 1949, Herbert and Poppe 1963, Johnson 1980, Comrie 1981, Kaun 1995). In Yowlumne, however, this is not the case. Instead, assimilation depends on the similarity between the trigger and target. Rounding applies only if both the trigger and target have the same vowel height. That is, round *high* vowels are followed by round high vowels, but by unrounded low vowels. Likewise, round *low* vowels are followed by round low vowels, but by unrounded high vowels. In (1), the forms with rounding harmony are marked with the symbol ' $\Box$ '.

(1) Yowlumne rounding harmony (Kuroda 1967:10,14)

1 <sup>31</sup> VOWEL IS ROUNDED			
HIGH	LOW		
mut-h <u>u</u> n ⊠	gop-hin □	шси	
'swear.AORIST'	'take care.AORIST'	пюп	2 <sup>ND</sup> VOWEL
mut-taw 🗆	gop-t <u>o</u> w ⊠	LOW	2 VOWEL
'swear.NDIR.GER'	'take care.NDIR.GER'	LOW	

The main characteristic of parasitic harmony is the mutual dependence of two features. This property is not limited to vowel harmony, but appears frequently in consonant harmony (Shaw 1991, Odden 1994, Gafos 1996/1999, Hansson 2001, Rose and Walker 2004, Mackenzie 2009). An example comes from Kalasha. Kalasha has retroflex harmony which is triggered by the leftmost coronal. Harmony applies if both the trigger and target have the same continuancy (2). That is, retroflex fricatives are followed by retroflex stops, but by non-retroflex stops. Similarly, retroflex stops are followed by retroflex stops, but by non-retroflex fricatives.

(2) Kalasha retroflex harmony (Arsenault and Kochetov to appear: ex. 7–12)

1 <sup>st</sup> coronal is retroflex					
FRICATIVE		STOP			
şu <u>ş</u> ik	Ø	tusu djek		FRICATIVE	
'to dry'		'to peck'		TRICATIVE	2 <sup>ND</sup> CORONAL
şit		t <sup>h</sup> e <u>t</u> karik	$\square$	STOP	2 CORONAL
'tight-fitting'		'to scatter'		3101	

This short presentation of the data alone strongly suggests that parasitic vowel harmony and consonant harmony can be integrated under a larger heading of parasitic assimilation. An assimilation pattern is parasitic if it depends on the similarity between a trigger and its targets. That is to say, parasitic assimilation occurs only when a trigger–target pair agrees in an otherwise independent feature, as in (3).

# (3) Parasitic assimilation

TRIC	GER		
GROUP A	GROUP B		
⊠ harmony	no harmony	GROUP A	TARGET
no harmony	☑ harmony	GROUP B	

Given these empirical similarities, one would expect the analyses of parasitic vowel harmony and consonant harmony to be formally similar. This was the case in the analyses based on feature geometry (Odden 1994). More recently, however, the two phenomena have received radically different treatment (Hansson 2001, Rose and Walker 2004).<sup>1</sup> The reason for such a wide theoretical gap lies in locality. In particular, most standard approaches to assimilation maintain strict locality (Ní Chiosáin and Padgett 2001, Gafos 1996/1999). The idea is that a particular feature can be realized over multiple targets only if such targets are adjacent to one another; no skipping is allowed.

There are at least two reasons why the locality facts of vowel and consonant harmony have been treated differently. The first lies in phonetics: vowel harmony has been shown to affect the articulation of transparent consonants and vowels (Boyce 1990, Benus 2005, Benus and Gafos 2007). The assumption was that consonant harmony has no such effects (Walker 2000, Rose 2004, Rose and Walker 2004). As it turns out, the coarticulatory effect has been since found in consonant harmony as well. In Kinyarwanda, for example, retroflex consonant harmony affects the articulator of intervening transparent consonants and vowels (Walker et al. 2008). The acoustic and articulatory data thus suggest that there is no phonetic basis for treating the two types of assimilation differently. If strict locality is observed, it must hold for vowel and consonant harmony alike.

The second reason to assume different mechanisms for vowel and consonant harmony has to do with the phonological facts. There are many cases of vowel harmony affecting all vowels, whereas consonant harmony invariantly affects a subset of consonants. This fact alone creates an illusion in which vowel harmony appears to be more local than consonant harmony. In particular, vowel harmony typically affects vowels in neighboring syllables, whereas consonant harmony affects consonants that may be several syllables away from one another.

In this paper, I argue that these different locality facts do not rule out a spreading account of consonant harmony. Since there is no a priori representational distinction between vowel and consonant harmony, I will argue that what sets them apart is an asymmetry within a class of OT constraints. This approach predicts that vowel harmony can be parasitic or not, whereas consonant harmony is always parasitic.

# 3. Non-parasitic assimilation

The different theoretical approaches to parasitic vowel harmony and consonant harmony appear to originate from some general property of assimilation. Thus, it seems reasonable to first look at the locality in non-parasitic assimilation and then compare it with parasitic assimilation. In particular, non-parasitic assimilation turns out to be the general pattern and parasitic assimilation a specific one. In OT terms, all kinds of assimilation are driven by some markedness constraint, whereas parasitic assimilation shows evidence for an additional constraint.

This section starts off with a typological overview of prominence and directionality in non-parasitic assimilation (section 3.1). These typological generalizations are attributed to a single class of markedness constraints—Licensed Alignment (section 3.2). The section concludes with a factorial typology of patterns predicted by Licensed Alignment constraints (section 3.3).

# 3.1. Prominence and directionality in non-parasitic assimilation

Phonological literature has attributed assimilation to several mechanisms. One idea is that features favor prominent positions. Another idea is that features prefer domain edges. This section reviews these two properties and focuses on the markedness constraints that underlie them.

Assimilation is often sensitive to particular positions. On the one hand, segments in prominent positions may act as triggers of assimilation. Such trigger prominence has been modeled with positional

<sup>&</sup>lt;sup>1</sup>Notable exceptions are Wayment 2009 and Nevins 2010.

faithfulness (Beckman 1997; 1998) in combination with a markedness constraint that drives assimilation (Baković 2000, Krämer 2003, Finley 2008, Mahanta 2008, Blaho 2008). On the other hand, prominent positions are often targets of assimilation. The most common approach that captures this second tendency is Positional Licensing (Steriade 1995, Zoll 1998a;b, Walker 2001; 2004; 2005; 2011, Kaplan 2008). The reasoning behind positional licensing is that a feature realized in prominent position has a different status than that feature realized in a non-prominent position. In terms of constraints, positional markedness constraints may require a feature to be associated with a morphologically, prosodically or psycholinguistically prominent position. One way to satisfy this condition is that the relevant feature is limited to a prominent position. Alternatively, a feature may target a prominent position. Consider the constraint in (4).

(4) LICENSE([high],  $\dot{\sigma}$ )

An output [high] must be associated with the stressed syllable.

The constraint LICENSE([high],  $\dot{\sigma}$ ) (4) can be satisfied by spreading the feature [high] from the a post-tonic segment to the stressed vowel.<sup>2</sup> Such a pattern is attested in Ascrea dialect of Italy (5).

(5)	Ascrea metapho	ny (Fanti 1938	; 1939; 1940	, Maiden 1991.	Walker 2005; 2011)
(-)			, ,	,	

a.	a <u>é</u> sse	'have.1SG.IMPF.SUBJ'	a <u>í</u> ∬i	'2sg.impf.subj'
	v <u>é</u> ∫te	'this.F.PL'	v <u>í</u> ∫tu	'M.SG'
	s <u>ó</u> rda	'deaf.F.SG'	s <u>ú</u> rdi	'M.PL'
	fj <u>ó</u> re	'flower.M.SG'	fj <u>ú</u> ri	'M.PL'
b.	t <u>ó</u> reua	'cloudy.F.SG'	t <u>ú</u> reuu	'M.SG'
	f <u>ó</u> sse	'be.3SG.IMPERF.SUBJ'	f <u>ú</u> sseru	'3pl.imperf.subj'

In Ascrea, stressed mid vowels are raised when followed by a high suffix vowel (5a). This happens even at a distance, and any intervening post-tonic vowels remain transparent (b).<sup>3</sup> The licensing constraint in (4) is satisfied by spreading to the stressed syllable (6b).

# (6) Licensing in Ascrea

/sórdi/	LICENSE([high], σ́)	Faith
a. sórdi	*!	
b. 🖙 súrdi		*

Yet the same licensing constraint is also satisfied by spreading from a pre-tonic segment (7a) or a segment in another word (7b). The challenge is that neither Ascrea nor any other language display such a pattern (Walker 2011).

- (7) Licensing in Ascrea fails (hypothetical)
  - a. Pre-tonic positions trigger metaphony

/isórda/	LICENSE([high], $\dot{\sigma}$ )	Faith
a. 😳 isórda	*!	
b. 🖙 isúrda		*

<sup>&</sup>lt;sup>2</sup>Throughout this paper, I assume privative features, but the current approach is entirely consistent with binary features (see Jurgec 2011b).

<sup>&</sup>lt;sup>3</sup>For expositional reasons, I abstract away from the patterns involving lax mid and low vowels. Additional constraints are required to capture these segments.

/sórda siné/	LICENSE([high], $\dot{\sigma}$ )	Faith
a. 🙂 sórda siné	*!	
b. 🖙 súrda siné		*

b. High vowels outside the word trigger metaphony

Positional licensing is generally not directional, predicting only bidirectional patterns. This clashes with the empirical data. While there are cases of vowel harmony triggered by stressed vowels and targeting preand post-tonic vowels (e.g. Pasiego height harmony), there are no reverse patterns, in which both pre- and post-tonic vowels target stressed vowels. Similarly, many cases of vowel harmony are triggered by a root vowel, targeting prefixes and suffixes (Baković 2000). However, the reverse pattern—in which both prefixes and suffixes affect roots—is unattested. Moreover, positional licensing does not refer to a domain, so there is no way to exclude distant triggers (even several words away). Such patterns have not been reported. In response, one way to proceed would be a redefinition of positional licensing to accommodate these pathologies. For example, the constraint could refer to particular non-prominent positions (e.g. post-tonic syllables in the case of Ascrea), but such a solution is not always available (e.g. the distinction between prefixes and suffixes). In short, the licensing constraint cannot be straightforwardly redefined to exclude these pathologies.

In contrast, the fact that assimilation is directional and domain-specific can be easily explained by another approach—Generalized Alignment (McCarthy and Prince 1993) that refers to segmental features.<sup>4</sup> Consider the constraint in (8).

(8) ALIGN([back], R; PWd, R; V) (Kirchner 1993, McCarthy 2003:78)
 ∀[back] if ∃PWd, assign one violation-mark ∀V that intervenes between Rightmost segment associated with [back] and the nearest Right Edge of some PWd.

In a word with multiple vocalic targets, the constraint ALIGN([back], R; PWd, R; V) (8) is satisfied as long as the final vowel within the word is [back]. To see the effect of this constraint, let us look at an example: Eastern Meadow Mari back harmony (9).

(9) Eastern Meadow Mari back narmony (vaysman 2009, walker 201	(9)	Eastern Meadow Mari back harmony (Vaysman 2	2009, Walker 2011	)
--	-----	---	-------------------	---

a.	á∫kəl-∫k <u>e</u>	'step-ILLAT.SG.NOM.POSS'
	wástər-∫k <u>e</u>	'maple-ILLAT.SG.NOM.POSS'
	køgørt∫én-ə∫k <u>e</u>	'dove-ILLAT.SG.NOM.POSS'
b.	kórno-∫k <u>o</u>	'bread-ILLAT.SG.NOM.POSS'
	ojləmá∫-ə∫k <u>o</u>	'story-ILLAT.SG.NOM.POSS'
	kuguʒán-ə∫k <u>o</u>	'princess-ILLAT.SG.NOM.POSS'
	kaβún-ə∫k <u>o</u>	'pumpkin-ILLAT.SG.NOM.POSS'

In Eastern Meadow Mari, words consisting of only front and central vowels may end on a front mid vowel (9a). A back vowel appearing anywhere in the root affects the final vowel (9b). This happens even at a distance, and the intervening central vowels remain unaffected.<sup>5</sup> The alignment constraint ALIGN([back], R; PWd, R; V) (8) is satisfied by spreading (10b).

<sup>&</sup>lt;sup>4</sup>Alignment is used by Kirchner (1993), Smolensky (1993), Cole and Kisseberth (1995), Itô and Mester (1995), Akinlabi (1996), Pulleyblank (1996), Golston (1996), McCarthy (1997), Ringen and Vago (1998), Archangeli and Pulleyblank (2002).

<sup>&</sup>lt;sup>5</sup>For expositional reasons, I abstract away from the patterns involving rounding harmony and high vowels, which need additional constraints.

(10) Alignment in Eastern Meadow Mari

/korno-∫ke/	ALIGN([back],R;PWd,R;V)	Faith
a. korno-∫ke	e!	
b. ☞ korno-∫ko		*

As it turns out, the constraint ALIGN([back],R; PWd, R; V) is satisfied by spreading to the very final vocalic target, whereas the intervening vowels do not matter. A low ranked faithfulness constraint prefers transparent non-final vowels (11b) over targets (11c).

(11) Alignment prefers transparency

/ojləmá∫ə-∫ke/	ALIGN-R([back],R;PWd,R;V)	FAITH	
a. ojləmá∫ə-∫ke	*!***		
b. ☞ ojləmáʃə-∫ko		*	
c. ojlomó∫o-∫ko		**!**	

In a word with multiple available vocalic targets, the alignment constraint is satisfied as long as the final vowel within the word is [back], whereas additional spreading to other vowels has no effect. Such a pattern is attested in Eastern Meadow Mari back harmony. However, back harmony without transparent vowels is much more common (e.g. Turkish). A low ranked faithfulness constraint will always prefer Mari to Turkish. The winning candidate in Turkish would be similar to (11c), which is harmonically bounded given the two constraints (Samek-Lodovici and Prince 1999; 2002). In short, the first problem of alignment is that it prefers transparency over targets (in combination with faithfulness). This is clearly a pathology, since transparent segments are generally more marked than targets. While it is possible to redefine alignment such that transparency would no longer be favored, the modified constraint would rule out the pattern found in Eastern Meadow Mari. The second challenge to alignment has to do with non-final targets when the final vowel is not subject to assimilation for other reasons (e.g. blocking). Assimilation often fails to reach the edge of the relevant domain. The definition of alignment in (8) solves this problem through gradience, assigning one violation mark for each non-target vowel. However, gradient constraints produce other pathologies (McCarthy 2003).

This section provided an overview of prominence and directionality in assimilation. The prominence effects can be captured by positional licensing, and directionality is predicted by alignment, but both approaches produce several pathologies.<sup>6</sup>

# 3.2. Licensed Alignment

In this section, I propose a new constraint—Licensed Alignment (henceforth, LA)—that can capture positional prominence and directionality in assimilation. At the same time, LA lacks the pathologies of positional licensing and alignment. Like licensing, LA refers to a particular (prominent) position within a larger domain. Yet LA is also sensitive to every instance of such a position. Like alignment, LA is directional, which means that it matters whether a potential target precedes or is preceded by a feature.

LA has been first proposed for stress by Hyde (2012), and this paper extends it to segmental features. LA constraints penalize triplets of a domain, a feature (F) and a third category ( $\gamma$ ), whenever F precedes  $\gamma$  within the domain (12).

<sup>&</sup>lt;sup>6</sup>The argument set forth in this section builds on Jurgec (2012b).

(12) \*domain[F,  $\gamma$ ]

a.

\* $\langle \text{domain}, F, \gamma \rangle$  / domain

F

b. Assign a violation for every triplet  $\langle domain, F, \gamma \rangle$ , when F precedes  $\gamma$  within the domain.

To illustrate how licensed alignment works, consider the constraint driving progressive nasal harmony (13). This constraint is violated once for each triplet  $\langle PWd, [nas], \times \rangle$  as long as the feature [nasal] precedes the root node (×), within the Prosodic Word.

(13) \*
$$\omega[nas, \times]$$
  
\* $\langle PWd, [nas], \times \rangle$  /  $PWd$   
[nas]  $\times$ 

Notice that LA refers to precedence among features and segments. The precedence relations can be made clear within an autosegmental view of assimilation (Goldsmith 1976; 1990, Clements 1976/1980; 1985a, Kiparsky 1981). In Autosegmental Phonology, assimilation is feature spreading, which involves adding association lines between a feature and a target root node. While precedence is typically defined for each tier (op. cit.) or at the segmental level (Raimy 2000), Hyde (2012) has shown that a definition spanning these levels is also possible. In particular, the constraint  $\omega[nas,\times](13)$  is satisfied by spreading because the target segment is associated with [nasal]. Two elements that are associated with one another are synchronous and do not precede each other. Let us examine these precedence relationships on an example.

Sundanese nasal harmony is triggered by a nasal sonorant  $\{m, n, p, \eta\}$  and applies rightwards until it encounters a consonant (Robins 1957).<sup>7</sup> The data is shown in (14). I follow the literature in that laryngeals also undergo nasal harmony (Cohn 1990). Other consonants never become nasal and block further assimilation, but this fact will not be analyzed here (see Walker 1998/2000 for a standard OT account based on feature co-occurrence constraints).

(14)	Nasal harmony in	Sundanese (Robins 1957:91,95)
	mãro	'to halve'
	pĩãr	'to seek'
	ŋũliat	'to stretch (INTR.)'
	kumãĥã	'how'
	ŋãjak	'to sift'
	mãwur	'to spread'

Nasal harmony in Sundanese can be attributed to the LA constraint  $\omega[nas, \times]$  (13). This constraint outranks the relevant faithfulness constraint. In what follows, I propose an autosegmental model of assimilation, hence the faithfulness constraint DEPLINK[F] will be used. In Sundanese, DEPLINK[nas] is active (15). This constraint is violated once for each segment linked to [nasal] in the output but not in the input.

(15) DEPLINK[nas] (cf. Itô et al. 1995, Myers 1997, Lombardi 1998, Morén 1999/2001, Blaho 2008) Let  $\times_i \Re \times_o$ .

Assign a violation mark, iff  $\times_o$  is associated with [nas] and  $\times_i$  is not.

Tableau (16) shows a ranking in which the LA constraint  $\omega[nas,\times]$  outranks the faithfulness constraint DEPLINK[nas]. The faithful candidate (a) incurs three violation marks of the LA constraint. This has to do with the fact that each of the segments following a [nasal] segment forms an offending triplet. LA constraints are categorical, because the locus of violation is a triplet, and each triplet can incur at most one violation mark. Spreading [nasal] to a target segment means that the feature [nasal] is synchronous with the

<sup>&</sup>lt;sup>7</sup>Sundanese is analyzed in Langendoen (1968), van der Hulst and Smith (1982), Cohn (1990; 1993), Piggott (1992), Piggott and van der Hulst (1997), Benua (1997), Walker and Pullum (1999), Walker (1998/2000).

segment. Hence, [nasal] no longer precedes that segment, and the relevant triplet no longer violates the LA constraint. Put differently, only oral segments following a nasal segment can violate the LA constraint, and the winning candidate (d) has no such segments.

[nas]		
/kumaha/	$\omega[nas,\times]$	DEPLINK[nas]
[nas]		
a. kumaha	$\langle \omega, [nas], a \rangle ! \langle \omega, [nas], h \rangle, \langle \omega, [nas], a \rangle$	
[nas]		
b. kumãha	$\langle \omega, [nas], h \rangle ! \langle \omega, [nas], a \rangle$	*
[nas]		
c. ku m ã ĥ a	$\langle \omega, [nas], a \rangle$ !	**
[nas]		
d. ☞ kumãĥã		***
[nas]		
e. kumahã	$\langle \omega, [nas], a \rangle ! \langle \omega, [nas], h \rangle$	*
[nas]		
f. kũ m ã hã		****!

Note that candidate (e) has two transparent segments. A transparent segment is preceded by a [nasal] segment, and this situation violates the LA constraint. At the same time,  $\omega$ [nasal,×] does not care that the transparent segment also *precedes* another [nasal] segment.

More generally speaking, triggers and targets are associated with the spreading feature, which is a way to express that the two segments are synchronous with the feature. Hence, triggers and targets will never be in a precedence relation with the spreading feature, satisfying the LA constraint. In contrast, all transparent segments will incur a violation mark of the relevant LA constraint. Put differently, transparent segments are inherently more marked than targets. This situation rules out the pathology seen in classical alignment constraints (11).

Recall that licensing constraints create two further pathologies (7). The first pathology is spreading from any direction. A closer look at candidate (16f) reveals that the rightward LA constraint cannot be satisfied by spreading in the leftward direction (and the low ranked faithfulness constraint inhibits such spreading). LA is intrinsically directional and will never prefer spreading in both directions. The second pathology of positional licensing is spreading outside of the domain. Because LA refers to a domain, spreading outside of that domain will never satisfy the constraint (but may create additional violation marks).

I have now shown that LA is successful at modeling assimilation, while lacking the pathologies of its predecessors. I now turn to some further predictions of LA.

# 3.3. Factorial Typology

kumāĥā 'how'

(16)

As we have seen, LA avoids many of the pathologies generated by the previous approaches. In this section, I show that LA can also distinguish between vocalic and consonantal targets. This ability is crucial in teasing apart the locality relationships in vowel and consonant harmony. More specifically, I show that the factorial typology of LA constraints excludes assimilation that prefers distant consonantal targets over closer vocalic targets. This prediction is in line with the empirical findings.

The LA constraint template refers to three categories: a feature, a domain and another category. This last category can be a root node, as we have seen in  $\omega[nas, \times]$  (13). Root nodes are targets in Sundanese,

yet one can easily imagine other kinds of targets. For example, the target could be a vowel (Turkish) or a stressed syllable (Ascrea). However, not all segments or their combinations are possible/attested targets in assimilation. For example, no case of rounding assimilation targets only coronals, voiced obstruents, or tense vowels. These patterns should be excluded by any restrictive theory of assimilation.

As it turns out, there is a principled way to restrict the last category of LA constraints. This is where the licensing part of LA comes into play. In section 3.1, I have already shown that assimilation may target prosodic/morphological domains rather than segments (Piggott and van der Hulst 1997, Hualde 1998, Piggott 2003, Walker 2011). That is to say, assimilation targets syllable nuclei, stressed syllables, or roots. This idea is central to Positional Licensing and LA.

I propose that the inventory of possible last categories/targets of LA constraints is restricted to prominent positions. One way to formalize this situation is through headedness: heads of prosodic or morphological domains are more prominent than other positions. Targets can be *prosodic heads* of syllables (Itô 1986/1988, Zec 1988/1994; 1995, Prince and Smolensky 1993/2004, Broselow et al. 1997, Morén 2000; 1999/2001, Gouskova 2004) or words (Kenstowicz 1997, de Lacy 2001; 2002; 2004; 2006; 2007, Gouskova 2010). When the domain is empty, all root nodes will be targeted, as in \* $\omega$ [nas,×] (13). Another option is that targets are *morphological heads* of words, which includes morphological heads of empty domains, that is all morphemes. The licensing restriction is stated in (17).

### (17) Licensing

For every Licensed Alignment constraint Cat1[Cat2, Cat3], if Cat1 is a domain and Cat2 is some [F], then Cat3 must be a head of some domain.

Let us look at these targets more closely. Prosodic targets can be used to capture many cases of assimilation. Consider that the targeted category of LA could be a syllable head (effectively, a vowel). The constraint  $\omega$ [vowel,ATR] favors spreading of [ATR] to all preceding vowels, as found in Turkana (Dimmendaal 1983, Vago and Leder 1987, Noske 1990; 1996; 2000). Targets can also be heads of prosodic words (stressed vowels). The constraint  $\omega$ [ $\delta$ ,high] prefers spreading of [high] from post-tonic to stressed vowels, as found in Ascrea metaphony (5).

Another option are morphological heads. For example, the target can be a root or any morpheme (i.e. a head of an empty morphological domain). Jurgec (2012a) builds on this idea and shows how LA constraints with morphological targets restrict assimilation only across morpheme boundaries.

Licensing (17) appears to be a significant, but independently grounded restriction on LA constraints. Furthermore, licensing has a direct prediction regarding locality in assimilation. All other things being equal, LA constraints prefer spreading to syllable nuclei, but not specifically to other syllable constituents. Vowels are better syllable nuclei then consonants, which means that spreading to vowels will be generally preferred over consonants. Simply put, licensing somewhat restricts assimilation with consonantal targets.

The fact that LA constraints favor vocalic targets can be demonstrated with a factorial typology. For any given feature [F], there is a set of LA constraints with different domains. If we limit the discussion to progressive spreading within a word, at least the following two constraints are possible:  $*\omega[F,\times]$  (18a) has the head of an empty domain as the third, targeted category, whereas  $*\omega[F,V]$  (18b) has the head of the syllable. Henceforth, 'vowel' will serve as a shorthand notation for the syllable head.<sup>8</sup>



These constraints exclude languages with only consonantal targets, regardless of the ranking. This is

<sup>&</sup>lt;sup>8</sup>There is a small difference between saying that targets are vowels or syllable heads. Consonants can be syllable heads, too, so the current approach predicts a language that has vowels and syllabic heads as targets. To the best of my knowledge, such a language has not been found. At the same time, there is no reported case of syllabic consonants behaving transparent just as regular consonants. Note that Positional Licensing makes identical predictions as LA with respect to syllabic consonants.

shown in (19), where we see spreading from the initial segment. Note that blocking is not considered here, as it needs separate constraints, and would distract from the main topic of this paper.

Let us first look at the evaluation of  $*\omega[F,\times]$ . This constraint is violated by triplets  $\langle \omega, [F], \times \rangle$ . Since all the candidates in (19) contain one prosodic word, one instance of [F], and four root nodes, this results in four different triplets. Each triplet differs from the others only in the final element, namely the root node. However, not all of these triplets violate the constraint. Consider candidate (d). We see that the LA constraint is violated only twice. Two of the triplets— $\langle \omega, [F], V_i \rangle$  and  $\langle \omega, [F], C \rangle$ —satisfy all the conditions in the constraint definition (18a): the root node must be preceded by [F], within a prosodic word. For example, the second root node is preceded by [F], because that root node is preceded by the initial root node, which is associated with [F]. The initial root node does not violate this constraint, since it is not preceded by [F]. This situation seems to be slightly different for the final root node, which appears to be preceded by the initial root node, which is itself associated with [F]. However, as we have already seen in Sundanese nasal harmony (16), association lines express synchrony between a root node and a feature, which then excludes any precedence relations. That is, the final root node is associated with [F], hence not preceded by [F]; the final root node is not part of a violating triplet. The same reasoning can be applied to the constraint  $*\omega[F,V]$ and all other candidates. Note that the two LA constraints are stringent. Any triplet that violates  $*\omega[F,V]$ also violates  $*\omega[F,\times]$ , but not vice versa.

$[F] \\ / \stackrel{ }{\times} V_i C V_j /$	*ω[F,V]	*ω[F,×]	DEPLINK[F]
$ \begin{array}{ c c c c } \hline & [F] \\ a. & \stackrel{ }{\times} & V_i \ C \ V_j \end{array} $	$\langle \omega, [F], V_i  angle, \langle \omega, [F], V_j  angle$	$\langle \omega, [F], V_i  angle, \langle \omega, [F], C  angle, \langle \omega, [F], V_j  angle$	     
$[F] b. \xrightarrow{[F]} V_i C V_j$	$\langle \omega, [{ m F}], { m V}_j  angle$	$\langle \omega,\![\mathrm{F}],\!\mathrm{C} angle,\!\langle \omega,\![\mathrm{F}],\!\mathrm{V}_{j} angle$	     * 
$[F] c. \mathbb{F} \times V_i C V_j$	$\langle \omega, [F], V_i \rangle, \langle \omega, [F], V_j \rangle!$	$\langle \omega,\! [{\rm F}],\! { m V}_i  angle,\! \langle \omega,\! [{\rm F}],\! { m V}_j  angle$	     * 
d. $\stackrel{[F]}{} V_i C V_j$	$\langle \omega, [F], V_i  angle$	$\langle \omega,\![{\rm F}],\!{ m V}_i angle,\!\langle \omega,\![{\rm F}],\!{ m C} angle$	     *
e. $\overset{[F]}{\underset{\times}{\overset{\downarrow}{V}_i C V_j}}$		$\langle \omega, [F], C  angle$	· · · **
$f. \overset{[F]}{\approx} \overset{\downarrow}{\times} \overset{V_i C V_j}{\times} V_i C V_j$	$\langle \omega, [{ m F}], { m V}_j  angle!$	$\langle \mathbf{\omega},\![\mathbf{F}],\!\mathbf{V}_j angle$	     ** 
$[F] \\ g \in \mathbb{F} \times V_i \subset V_j$	$\langle \omega, [{ m F}], { m V}_i  angle!$	$\langle \omega, [{ m F}], { m V}_i  angle$	     **
h. $\overset{[F]}{\times}$ V <sub>i</sub> C V <sub>j</sub>		_	     ***

(19) Spreading to consonants across vowels ruled out

Returning to the factorial typology, what (19) shows is that some candidates (marked with " $\mathfrak{P}$ ") are harmonically bounded. First, candidate (c) has spreading to the consonant, while the intervening vowel is transparent. This candidate incurs one more violation mark of  $\mathfrak{P}$ [F,V] compared to candidates (b) and (d) in which only vowels are targeted, without faring better on any other constraint. Candidates (f) and (g) have spreading to the consonant and one vowel, but not to the other vowel. The candidates should be understood as both having transparent vowels but no transparent consonants. Such patterns are not attested, and are ruled out by the factorial typology based on LA and faithfulness constraints. All other candidates are winners under some ranking.<sup>9</sup>

We have now seen that the LA constraints rule out all patterns in which vowels are transparent but consonants are not. This is exactly what is required in an analysis of non-parasitic assimilation. In other words, two types of assimilation are possible: one that targets vowels and the other that targets all segments.

The asymmetry between consonantal and vocalic targets has been observed in the literature (Howard 1972, Jensen 1974, Clements 1985b; 1991, Sagey 1990, Steriade 1995, Morén 2003). However, this generalization is typically couched in terms of blocking, not transparency. The two main observations are: (i) vowel place features often spread from one vowel to another across the intervening consonants, and (ii) consonant major place features never spread from one consonant to another (Hansson 2001). The problem with this generalization is that both types of place features do spread among vowels and consonants, as long as they are adjacent (Odden 1994). An alternative view of this situation is that transparency is implicational: vowels can be transparent only if consonants are also transparent. Thus, spreading of a feature across vowels is restricted. In contrast, spreading of any feature to adjacent segments is possible, because transparency is not involved.

The crucial comparison is between vowel and consonant harmony. Vowel harmony is a common pattern which typically affects only vowels, with consonants being transparent. Consonant harmony is another common pattern, which often affects only consonants. However, no known case of consonant harmony affects all consonants without also affecting vowels, and all known cases of consonant harmony affect only a subset of consonants.<sup>10</sup>

To summarize, the patterns with transparent segments reveal an asymmetry between vocalic and consonantal targets. Vocalic targets are generally preferred to consonantal targets. This means that a consonant will generally assimilate only if intervening vowels do, too. The remaining question is how we can capture the attested consonant harmony cases. My proposal is that consonant harmony is directly parallel to parasitic vowel harmony: both involve assimilation within a small subset of segments. As I will show, parasitic vowel harmony affects only a subset of vowels, whereas consonant harmony affects only a subset of consonants. This effect will be attributed to a high ranked constraint that prefers similarity between triggers and targets.

With this in mind, we can now proceed to the analysis of parasitic assimilation in Yowlumne and Kalasha. It will be shown that Yowlumne requires an LA constraint targeting vowels, while Kalasha requires an LA constraint targeting all segments. At the same time, both patterns require an additional markedness constraint.

### 4. Yowlumne

I now proceed to the analysis of parasitic vowel harmony in Yowlumne. I first show that the constraints proposed for non-parasitic assimilation are not sufficient to account for parasitic assimilation (section 4.1). Next, I propose an additional agreement constraint required for parasitic assimilation (section 4.2). Finally, I give an analysis of Yowlumne based on alignment and agreement (section 4.3).

<sup>&</sup>lt;sup>9</sup>A reviewer notes that additional constraints could produce the unattested pattern with only consonantal targets (19c). Consider a high ranked feature co-occurrence constraint against the combination of the spreading feature and a vowel. This constraint would prefer spreading to consonants alone in combination with a lower ranked LA constraint  $*\omega[F,\times]$ . The challenge at hand is that feature co-occurrence constraints favor transparent segments over targets. This is clearly a pathology, since transparent segments are intrinsically more marked than targets. Put differently, feature co-occurrence constraints themselves make pathological predictions, which need to be addressed independently of the current account based on LA constraints. In Jurgec (2011a), I propose that transparent segments violate feature co-occurrence constraints just as targets do. Hence, feature co-occurrence constraints cannot prefer some targets over others (e.g. consonants over vowels).

<sup>&</sup>lt;sup>10</sup>To the best of my knowledge, only one pattern comes close to such an unattested pattern: Karaim palatalization consonant harmony (Kowalski 1929, Hamp 1976, Nevins and Vaux 2004, Hansson 2007b, Nevins 2010). All consonants are affected, with two exceptions. First, the glide [j] is not affected and can appear in non-palatalized environments. Second, this process also targets the high back vowel. Hence, Karaim does not constitute a clear counter-example.

#### 4.1. Licensed Alignment is not enough

The Yowlumne dialect of Yokuts is likely the most widely discussed case of parasitic vowel harmony.<sup>11</sup> As we have seen in (1), Yowlumne has rounding harmony which targets  $\{i, a\}$ , which turn into  $\{u, o\}$ . However, rounding is found only when the trigger and the targets agree in vowel height. More data are provided in (20).<sup>12</sup>

(20)	Yowlumne vowel harmony (Kuroda 1967:10,14)			
	gij'-hin	gij'-taw	'touch'	
	mut-h <u>u</u> n ⊄	mut-taw	'swear'	
	xat-hin	xat-taw	'eat'	
	gop-hin	gop-t <u>o</u> w ∅	'take care of an infant'	
	'AORIST'	'NDIR.GER'		

Parasitic vowel harmony is much less frequent than non-parastic vowel harmony, and not all types are attested. Rounding harmony, for example, can be parasitic on height and backness, but not on other features (Kaun 1995). Given these gaps, it seems reasonable to ask if the current approach restricts parasitism in any way. While it is in principle possible to restrict the parasitic feature, this task nevertheless falls outside this paper. In no way is this challenge specific to the current approach; the alternatives make no or few principled restrictions. The aim here is to demonstrate that parasitic assimilation can be analyzed as spreading and OT constraints.

Winner $\sim$ Loser	*[rd high]	*[rd low]	*@[rd,V]	$*\omega[rd, \times]$	DEPLINK[rd]
[rd] [rd] m u t'hun~m u t'hin	L	     	W	W	L
$\begin{bmatrix} [rd] & [rd] \\ m & t'taw \sim m & t'tow \end{bmatrix}$		W	L	L	W
$\begin{bmatrix} rd \\ g & p & to & w \\ e & y & b & ta \\ g & y & ta & w \\ \end{bmatrix}$		L	W	W	L
[rd] [rd] g o phin~g o phun	w	     	L	L	W

(21) Inconsistency in Yowlumne

In line with the previous analyses, I take [round] as the spreading feature. Licensing (17) predicts at least two LA constraints with [round]:  $*\omega$ [round,V] and  $*\omega$ [round,×]. Another potentially relevant constraint is the faithfulness constraint DEPLINK[round]. However, since spreading to vowels is restricted depending on vowel height, we can easily imagine other potentially active constraints. In particular, feature co-occurrence constraints have been shown to be required for assimilation (Archangeli and Pulleyblank 1994, Walker 1998/2000), and in Yowlumne, two such constraints may be involved: \*[round high] and \*[round low] (Kaun 1995; 2004). The feature constraints may block assimilation depending on the target vowel

<sup>&</sup>lt;sup>11</sup>The original data come from Newman (1944). Subsequent treatments include Kuroda (1967), Kisseberth (1969), Gamble (1978), Jensen and Stong-Jensen (1979), Archangeli (1984; 1985; 1988), Archangeli and Pulleyblank (1994), Noske (1985), Steriade (1986), Cole (1987), Cole and Trigo (1988), Zoll (1993), Goldsmith (1993), Noske (1993), Kaun (1995), Sprouse (1997), Archangeli and Suzuki (1997), McCarthy (1999; 2007b), Orgun and Sprouse (2007), Dresher (2009), Nevins (2010).

<sup>&</sup>lt;sup>12</sup>Note that vowel harmony applies rightwards and iteratively (cf. [t'it'-ijin] 'raccoon' ~ [tuk'-<u>uju</u>n] 'jackrabbit'). Some suffixes are underlyingly rounded themselves (cf. [t'aw-hatin-x<u>oo</u>-hin] 'was trying to win'), which is enforced by high ranked MAXLINK[round].

height. As we can see in comparative tableau (21), all these constraints are unable to capture Yowlumne. In particular, since no constraint prefers only winners, there is no consistent ranking possible to capture the Yowlumne pattern.

The alignment, faithfulness and feature co-occurrence constraints fail to capture parasitic assimilation. In response, I propose another class of markedness constraints.

# 4.2. Agreement

In this section, I argue that the constraint required for parasitic assimilation is agreement. I first show that the classic definition of AGREE faces several formal and typological challenges. As a solution, I propose a new kind of agreement constraint. In a nutshell, the idea is that agreement works in combination with licensed alignment.

Classic agreement constraints prefer similar segments. Let us look at one such constraint, AGREE[voice] in (22).

# (22) AGREE[voice]

- a. Adjacent segments must have the same value of the feature [voice]. (Baković 2000:4)
- b. Obstruent clusters must agree in voicing. (Lombardi 1999:272)
- c. A segment has [voice] iff its neighboring segments have [voice]. (Blaho 2008:64,139)

The constraint AGREE[voice] has several definitions. However, all the definitions in (22) include reference to three variables: one feature and two segments at some distance from one another. More specifically, the constraint AGREE[voice] is violated when two segments are adjacent to one another and only one is associated with [voice]. This seems to be ideal to capture voicing assimilation, since two adjacent obstruents with the same voicing satisfy AGREE[voice], whereas two obstruents with different voicing violate the constraint. Thus, it seems that classic agreement constraints correctly predict at least some types of assimilation.

Another potential advantage is in the adjacency requirement that is ingrained in the constraint definition (22). No known language has voicing assimilation of a word-initial obstruent targeted by a word-final one. Agreement constraints cannot generate such a pattern, since any two non-adjacent segments vacuously satisfy the constraint. That is to say, no pair of non-adjacent obstruents can violate AGREE[voice]. However, while adjacency is undoubtedly an important concept in phonology (Howard 1972, Jensen 1974, Odden 1994, Halle 1995), its reference in a constraint that prefers assimilation creates pathologies. For example, agreement constraints are equally satisfied by assimilation or epenthesis. The choice between the two can be determined by other constraints. Tableau (23) shows a ranking that prefers epenthesis, as in candidate (c).

	apga	AGREE[voice]	DEPLINK[voice]	Dep
a.	apga	*!		
b.	abga		*!	
c. 🖾	° apiga			*

(23) Agreement drives epenthesis (unattested)

The grammar in (23) has epenthesis only when obstruents disagree in voicing, but not otherwise. No such cases have been reported, and their existence would be surprising. This kind of pathology is an instantiation of the too-many-solutions problem (Pater 1999, Wilson 2000; 2001, Steriade 2001, 2001/2008; Blumenfeld 2006, Baković 2007). One way to solve this challenge is a modification of agreement constraints.

The question is how to amend agreement in such a way that it would no longer produce pathologies. To start with, let us look at the definitions of agreement constraints in (22). All these definitions refer to

one feature and adjacent segments. Upon closer examination, however, it becomes clear that reference to two features is required. In particular, only obstruents can potentially violate the constraint, while other segments are ignored.

I propose two modifications of agreement constraints. First, agreement should explicitly refer to two features rather than one. Second, adjacency should be dropped completely from the definition. The reason why this is possible in the current approach is that locality is already established by alignment constraints, which I intend to use together with agreement. By modifying Blaho's definition in (22c), the new agreement constraint template is in (24).

# (24) AGREE[F,G]

A root node is associated with [F] and [G] iff all root nodes associated with [F] are also associated with some [G].

The constraint in (24) contains reference to two features, which are not equivalent with respect to their role in the definition. In particular, AGREE[F,G] is not identical to AGREE[G,F]. There are other ways in which the newly defined agreement constraint is very different from classic agreement. For example, the new agreement is vacuously satisfied by no spreading or by spreading within segments that lack one of the two features. To make the differences more explicit, let us look at evaluation of AGREE[F,G] in (25).

(25)	AGREE[F,G] evaluations
------	------------------------

AGKEE[I,O] C	valuations
	AGREE[F,G]
[F]	
$\begin{vmatrix} \dot{\mathbf{x}} & \dot{\mathbf{x}} \\ \mathbf{a} & \mathbf{G} \end{bmatrix}$	
b. $\times \times \times$	(v.s.)
[F]	
$\times_i \times_j$	
c. [G]	$\times_i$
[F]	
$\times_i \times_j$	
d. [G]	$ imes_j$
[F]	
e. [G]	
[F]	
f. [G] [G]	
[F]	
$g \cdot \overset{I}{\times} \times$	(v.s.)
[F]	
× ×	
h. [Ġ]	
[F]	
i. [G]	(v.s.)

Candidates (a–f) have spreading. In candidate (a), both root nodes are associated with [G]. Hence, the agreement constraint is satisfied. Candidate (b) vacuously satisfies the constraint since no root node is associated with [G]. In contrast, candidates (c) and (d) have only one root node associated with [G]. These

two candidates violate AGREE[F,G], because they have two root nodes that are associated with the same instance of [F], yet only one of the two root nodes is associated with [G]. In other words, it is not the case that all root nodes associated with [F] are also associated with some [G]. Directionality has no effect here. So far then, the constraint seems to be doing its job, which is enforcing spreading of [F] within segments that are linked to some [G].

The candidates considered next reveal two further properties of agreement constraints. Candidate (e) has spreading of [G] rather than [F]. This candidate does not violate AGREE[F,G], because only one root node is linked to [F]. However, if we compare candidate (e) with candidate (i) without spreading, we see that they do not differ in terms of AGREE[F,G], which means that this constraint alone cannot enforce spreading of [G]. Candidate (f) contains a transparent segment. The constraint AGREE[F,G] is not violated by this candidate, because the constraint can see only segments associated with [F], while the rest are ignored. The initial and final root nodes are both associated with [F] and some [G], satisfying agreement.

Finally, candidates (g–i) have no spreading. Regardless of whether the segment associated with [F] is also associated with [G] or not, these candidates all satisfy the constraint AGREE[F,G]. At first, this seems a bit counterintuitive. In particular, what we see here is that AGREE[F,G] can only be violated by candidates with spreading. Furthermore, the faithfulness constraint DEPLINK[F] prefers no spreading, regardless of its ranking. However, we already have a constraint that prefers spreading—licensed alignment. A combination of an LA constraint and a higher ranked agreement constraint prefers spreading of [F] only if the target agrees with the trigger with respect to [G]. Put differently, the role of agreement constraints is to restrict spreading to a subset of targets of the LA constraint.

As we have now seen, the new definition of agreement differs substantially from its predecessors. First, the new agreement constraint does not refer to adjacency of segments, which rules out the pathologies generated by classical agreement. Second, the new agreement does not trigger spreading on its own, but rather restricts spreading to the targets similar to the trigger. In this sense, agreement somewhat resembles constraints in an alternative approach to consonant harmony (Agreement by Correspondence, see section 6 below).

The question is whether we see effects of agreement outside parasitic assimilation. As it turns out, similar constraints have been proposed for prosody. Myrberg (2010) uses a constraint \*ADJUNCTION which penalizes sister nodes in the prosodic structure that are not instantiations of the same prosodic category. Selkirk (2011) and Elfner (2011) propose a directional variant of the same constraint. Although these constraints refer to prosodic structures, they embody the same idea, namely agreement between two constituents that are linked to another (higher) constituent.

To summarize, the new agreement constraint is violated solely by spreading to targets that differ from triggers. This is exactly what is required in an analysis of Yowlumne, in which rounding never spreads to the vowel which has different vowel height that the trigger. The following section gives an account of parasitic harmony in Yowlumne, which demonstrates that agreement and alignment work hand in hand.

# 4.3. Analysis

Yowlumne parasitic vowel harmony requires three constraints. As we have seen above, alignment alone is not sufficient to capture the parasitic pattern. A high ranked agreement constraint (26) is necessary in addition. The constraint AGREE[round,high] penalizes any instantiation of [round] linked to high and low segments at the same time.

#### (26) AGREE[rd,hi]

A root node is associated with [round] and [high] iff all root nodes associated with [round] are also associated with some [high].

I will show the effect of these constraints in three steps. First, I will consider a combination of two high vowels. Next is the input with two low vowels. In both cases, rounding harmony applies. The final input is

one with a high trigger followed by a low target. As we will see, the agreement constraint can be violated only in this last case, preferring no harmony.

In (27), we see an input with two high vowels. When vowel height is preserved, AGREE[rd,hi] cannot be violated. Alignment is ranked next, and the spreading candidate (b) fares best on it, winning over the faithful candidate (a).

(27)	AGREE satisfied, ALIGN pre	fers harmony		
	[rd]			
	/ m u t' - h i n / [hi] [hi]	AGREE[rd,hi]	*ω[rd,V]	DepLink[rd]
	[rd] muťhin		*1	
	a. [hi] [hi]		···!	
	m u t'h u n b. ☞ ma ma			*

Tableau (28) shows an input with two low vowels. Agreement is vacuously satisfied by all outputs that lack high (or round) segments, as it is the case for both candidates. LA is ranked next, and it prefers the candidate with rounding harmony (b). In short, harmony is preferred whenever the targets are of the same height as the trigger.

(28) AGREE vacuously satisfied, ALIGN prefers narmony
---

[rd]			
/ g o p - t a w /			
	AGREE[rd,hi]	*ω[rd,V]	DEPLINK[rd]
[rd]			
g o pt a w			
a. [lo] [lo]		*!	
[rd]			
g o p t o w			
b. ☞ [lo] [lo]			*

(29)

AGREE violated by spreading

[rd]			
/muť-taw/			
[hi] $[lo]$	AGREE[rd,hi]	*ω[rd,V]	DEPLINK[rd]
[rd]			
mut'taw a.☞ [hi] [lo]		*	
[rd]			
b. [hi] [lo]	*!		*

Finally, the input in (29) has two different vowel heights. It is only in this case that agreement becomes active. In particular, spreading violates AGREE[round,high] since [round] is associated to some segments that are [high] and some that are not. Hence, the faithful candidate (a) wins over the spreading candidate (b).

Yowlumne shows the joint effect of LA and agreement. LA prefers spreading to all segments or to vowels, whereas agreement prefers spreading to a subset of those targets. Put differently, LA targets some segments, while it remains agnostic about the triggers. Agreement, on the other hand, prefers spreading only when the trigger is identical to the target in terms of another feature.

### 5. Kalasha

In this section, I show that Kalasha retroflex harmony works very much like Yowlumne vowel harmony. The two patterns are much alike: both are parasitic. In terms of the analysis, the parasitic effect is attributed to agreement constraints. At the same time, there are two crucial differences in the analysis of Kalasha and Yowlumne. First, the LA constraint in Kalasha targets all root nodes rather than vowels as in Yowlumne. Second, Kalasha is multiply parasitic and shows evidence for several agreement constraints, whereas Yowlumne requires only one.

# 5.1. Coronal harmony

Kalasha (Morgenstierne 1973, Trail n.d., Mørch and Heegård 1997, Heegård and Mørch 2004, Bashir 2003, Arsenault and Kochetov to appear, Arsenault 2012) exhibits restrictions on retroflexion that strongly resemble the Yowlumne pattern. Retroflexion of coronals within roots is restricted, such that any two coronal stops are either both retroflex or neither is. This generalization is true for stops and fricatives, as seen in (2). In addition, affricates behave the same way, as shown below in (30). Kalasha consonant harmony turns out to be a three-way parasitic pattern, distinguishing between stops, fricatives, and affricates. One way to interpret the data in (30) is to say that the trigger is the initial coronal, whereas the target is the following consonant. This alternation is limited to roots, and there are no active alternations to support retroflex harmony. However, Richness of the Base allows for a combined treatment of alternations, and in this section I will show that it can also model static restrictions within morphemes.

	1 <sup>51</sup> CORONAL			
FRICATIVE	AFFRICATE	STOP		
<u>şuş</u> ik ⊠ 'to dry'	(no data)	tusu djek 'to peck'	FRIC	NAL
şatç	dzats ∅	tõtçuk	FR	RO
'shelter'	'spirit beings'	'active'	AF	CO
şit	tşat	<u>t</u> het karik ⊠	OP	ND
'tight-fitting'	'moment'	'to scatter'	ST	2

(30) Kalasha retroflex harmony (Arsenault and Kochetov to appear: ex. 7–12)

Kalasha is a case of retroflex harmony. One way to analyze this pattern is to say that it involves two binary features, [-anterior] and [-distributed] (e.g. Johnson 1972, Schein and Steriade 1986). A privative alternative is the feature [retroflex]. The feature [retroflex] may be a dependent on the feature [coronal], which is in line with most previous accounts (Schein and Steriade 1986, Sagey 1990, Hall 1997, Ní Chiosáin and Padgett 2001, Hansson 2001, Rose and Walker 2004), but the present analysis based on LA constraints does not require such a representational restriction. I use [retroflex] here for expositional reasons, and without precluding an analysis assuming other features.

The LA constraint active in Kalasha could require spreading of [retroflex] to either vowels or all root nodes, as required by licensing (17). Recall that in the analysis of Yowlumne, vowels are a possible third category of the relevant LA constraint. In contrast, Kalasha retroflex harmony targets only consonants, hence vowels cannot be the third category of the active LA constraint. Instead, the third category has to be the root node. This is further supported by the fact that vowels may also be retroflex as in Badaga (Ladefoged and Maddieson 1996) or some dialects of Kalasha (Mørch and Heegård 1997, Heegård and Mørch 2004).<sup>13</sup> The domain of the constraint is the root (31).

(31) \*root[retroflex,×] \* $\langle root, [rx], \times \rangle$  / root [rx] ×

Several agreement constraints are required in Kalasha. The first agreement constraint requires that the trigger and targets agree in the feature [coronal] (32). Another assures that only obstruents are affected (33). Combined, retroflex harmony is limited to coronal obstruents.

(32) AGREE[rx,cor]

A root node is associated with [retroflex] and [coronal] iff all root nodes associated with [retroflex] are also associated with some [coronal].

(33) AGREE[rx,son]

A root node is associated with [retroflex] and [sonorant] iff all root nodes associated with [retroflex] are also associated with some [sonorant].

The last is a faithfulness constraint DEPLINK[retroflex]. This constraint is ranked below LA, parallel to the ranking in Yowlumne.

I first show the ranking that limits spreading to coronal obstruents. Tableau (34) illustrates the effect of high ranked agreement constraints in combination with LA. First, LA generally prefers spreading to all segments, as in candidate (d). Transparent segments violate the constraint since they satisfy the requirement, which is that a [retroflex] segment precedes them, while they are not associated to that instance of [retroflex]. However, candidate (d) with total harmony violates agreement constraints. Only candidates (a) and (b) do not. Of these, spreading to a single coronal obstruent as in candidate (a) fares best on alignment. In short, agreement constraints restrict spreading to a subset of targets determined by the LA constraint.

Many cases of consonant harmony are similar to the hypothetical example in (34). That is, consonant harmony applies within a class of coronals, and there is no additional parasitic requirement. For example, Aari has sibilant harmony which applies to fricatives *and* affricates, even if in the same word (Hayward 1988; 1990). Other similar examples can be found in Koyra (Hayward 1982; 1988), Chumash (Applegate 1972, Beeler 1970, Harrington 1974, Poser 1982; 1993; 2004, Lieber 1987, Gafos 1996, Hansson 2001, McCarthy 2007a), and Slovenian (Jurgec 2011a) sibilant harmony. This shows that all kinds of coronal harmony can be captured using this approach, namely by using a combination of agreement and alignment constraints. For an extension to other types of consonant harmony see Jurgec (2011a).

<sup>&</sup>lt;sup>13</sup>Mørch and Heegård (1997), Heegård and Mørch (2004) describe the language as having retroflex vowels, which variantly spread to other segments. This fact is not specifically analyzed in Arsenault and Kochetov (to appear). From the existing literature, it is impossible to tell whether and how the two patterns interact. If they do, then Kalasha does not exhibit parasitic retroflex harmony, but simply local spreading. Henceforth, I will assume that the data of Arsenault and Kochetov (to appear) is correct and that vowel retroflexion is a separate process, which possibly interacts with consonant retroflexion.

(34) $/[tatnak] \rightarrow [tatnak] (hypothetical)$	)			
[rx]		1		
/ t a t n a k/		1		
		1		
		l I		
[son] [son]	AGREE[rx,cor]	AGREE[rx,son]	*rt[rx,×]	DEPLINK[rx]
[rx]		l I		
t a t n a k [cor] [cor] [cor]		       		
a. [son] [son] [son]		   	*****!	
[rx]		   		
tat nya k				
[cor] [cor] [cor]				
b. 🖙 [son] [son] [son]		   	****	*
[rx]		 		
t a t m a k		1		
		1		
[cor] [cor] [cor]		-   		
			***	**
c. [son] [son] [son]		<u>η</u> !	* * *	~ ~
[rx]		 		
$t a^{-} t \eta a^{-} k^{-}$		-   		
[cor] [cor] [cor]		1		
		1		
d. [son] [son]	t(!)tŋ	a~(!)ηi~		****
		l		

# 5.2. Multiple parasitism

Kalasha retroflex harmony is multiply parasitic. First, it applies only to coronal obstruents. Second, it applies only if the trigger and the target are both stops, both fricatives, or both affricates. This second type of parasitism will be analyzed now.

There are several ways to distinguish between stops, fricatives, and affricates in terms of features and consequently constraints that refer to them. In addition to the machinery developed in the previous section, I propose two other features and their agreement constraints. The first feature is [continuant], which captures the difference between fricatives and the other two groups. The second feature is [delayed release], and this feature singles out affricates. (Other analysis are also possible, for example a feature that refers to stops exclusively.) The two features are referred to by two agreement constraints: AGREE[retroflex, continuant] and AGREE[retroflex, delayed release]. I follow Arsenault and Kochetov (to appear) in assuming that Kalasha has progressive spreading determined by the coronal in the initial syllable.<sup>14</sup>

Tableau (35) shows a root with two fricatives. In this case, the winning candidate (b) has spreading

<sup>&</sup>lt;sup>14</sup>This can be captured by using positional faithfulness to leftmost coronal (Beckman 1997; 1998, Barnes 2006, Jurgec 2011a), which will not be analyzed further. In this sense, the current analysis is directly parallel to the alternatives to be discussed in section 6.

to the second fricative, as opposed to the faithful candidate (a). While spreading to all root nodes (c) would perfectly satisfy alignment, such a candidate incurs several violation marks of AGREE[retroflex, sonorant] and AGREE[retroflex, continuant]. This situation is directly parallel to what we have seen in tableau (34). Furthermore, the current analysis resembles Yowlumne in that harmony within coronal fricatives is preferred by LA constraints, while agreement constraints are not violated by such candidates.



Next, let us consider inputs containing two kinds of coronal obstruents. Tableau (36) shows an input with a retroflex fricative followed by a stop. In this case, alignment would prefer spreading, as in candidate (b). Spreading, however, violates agreement, and the faithful candidate (a) is preferred instead.

(36) AGREE prefers similarity

TOKEE protors similarity				
$[rx]$ $/ \begin{array}{c} [rx] \\ i \\ [son] \\ [cont] \\ [con$	AGREE [rx,son] AGREE	[rx,cont] AGREE [rx,d.r]	*rt[rx,×]	DEPLINK[rx]
			**	
a. <sup>Log</sup> [cont] [cont]		   		
s i t [son]		       	*	*
$[\text{cont}]^{\text{cont}}$				

Tableau (37) has an input containing an affricate and a stop. Directly parallel to the situation in (36), agreement constraints inhibit spreading in such cases. The faithful candidate (a) wins regardless of its violations of alignment.

AGREE prefers similarity					
$ \begin{bmatrix} rx \\   \\ / ts & a & t / \\   \\ d.r \end{bmatrix} $	AGREE [rx.cont] AGREE [rx.d.r] *rt[rx ×]	DEPLINK[rx]			
$\begin{bmatrix} [rx] \\ & t \\ ts & a & t \\ & [d,r] \\ a.  cont \end{bmatrix}$		*			
$\begin{bmatrix} [rx] \\ \\ ts \\ a \\ t \\ [d,r] \\ b.  [cont] \end{bmatrix}$	· *i *	* *			

We have now seen that the current approach based on LA and high ranked agreement can successfully account for both types of parasitic assimilation. The two analyzed languages serve as case studies. This approach can be extended to other cases of consonant and parasitic vowel harmony. Furthermore, LA takes care of non-parasitic assimilation and its preference for vocalic rather than consonantal targets. This reveals a crucial difference between the two patterns. Since LA never prefers spreading to consonants alone, vowel harmony can be parasitic or not, whereas consonant harmony is always parasitic.

# 6. Alternatives

(37)

The main contribution of this paper is to show that parasitic vowel harmony and consonant harmony are alike and have a similar spreading analysis enforced by OT constraints. Crucially, a spreading account of consonant harmony is entirely possible (contra Walker 2000, Hansson 2001, Rose 2004, Rose and Walker 2004, Arsenault and Kochetov to appear, Hansson 2010, Nevins 2010). Now I shortly discuss one prominent recent alternative—Agreement by Correspondence (Hansson 2001, Rose and Walker 2004, et seq.; henceforth, ABC).

ABC is an approach to assimilation that has been developed specifically for consonant harmony. The main idea is that consonant harmony is due to long-distance consonant agreement, not spreading. This is formalized in terms of output correspondence relations between triggers and targets. ABC can certainly deal with all kinds of consonant harmony and has even been extended to model parasitic vowel harmony (Rhodes 2011). My argument here is not that ABC fails to capture parasitic assimilation. Instead, I claim that ABC says nothing about non-parasitic assimilation, which is problematic per se. To model non-parasitic assimilation, ABC relies on constraints and representations that are traditionally employed in such cases. The current proposal unifies both phenomena under a single analysis without any reference to correspondence. In short, ABC is less parsimonious.

Occam's Razor is sufficient to make ABC suspect. However, there are several other arguments why it is less capable of capturing assimilation than the current proposal. First, ABC restricts blocking. Blocking is arguably rare in consonant harmony (Hansson 2001:213,214; Rose and Walker 2004:486,487), but it is attested. Examples include Sanskrit (Johnson 1972, Selkirk 1980, Kiparsky 1985, Schein and Steriade 1986,

Cho 1991, Hall 1997, Ní Chiosáin and Padgett 1997, Gafos 1996/1999, Hansson 2001, Rose and Walker 2004, Kaplan 2008), Kinyarwanda (Walker et al. 2008), Berber (Hansson 2010), and Slovenian (Jurgec 2011a). Hansson (2007a) shows that correspondence constraints can model blocking in an ABC-based approach. The idea is that when a high ranked constraint (e.g. a faithfulness or feature co-occurrence constraint) prohibits the blocker from undergoing assimilation. Since correspondence looks at all segments of a particular kind, this can include the blocker, which in turn could violate identity constraints that apply among the corresponding consonants. As a result, blocking is preferred over transparency. However, the predictions are less clear when non-parasitic assimilation is concerned. In particular, ABC excludes blockers that are different from the trigger–target pair. For example, ABC excludes blocking of vowel harmony by a subset of consonants. The problem is that such blocking is found in many unrelated languages, including Turkish (Clements and Sezer 1982), Finnish (Kiparsky 1981), Shona (Uffmann 2006), Nawuri (Casali 1995), Buchan Scots (Paster 2004), Akan (Clements 1976/1980; 1985a), Turkana (Dimmendaal 1983, Noske 1990; 2000), Assamese (Mahanta 2008), Ainu (Itô 1984), and Yucatec Maya (Krämer 2001).<sup>15</sup>

The second argument concerns the trigger-target disparities. Assimilation is sometimes triggered by a segment that is *different* from its targets. For example, a consonant may affect a vowel but not other consonants—as in Serbo-Croatian palatalization (Morén 2003). This can happen even if a trigger and its targets are not adjacent. For example, faucal harmony in Snchitsu?umshtsn is triggered by a set of consonants that target vowels (Bessell 1998). Similarly, progressive emphasis spread in Northern Palestinian Arabic (Davis 1995) is triggered by consonants and targets the low vowel across other consonants. Chilcotin flattening is triggered by consonants that target vowels, whether adjacent or not (Cook 1976; 1983; 1987; 1993). In ABC, such patterns would be treated as long-distance rather than locally applying. The challenge is that ABC assumes that all long-distance assimilations must be parasitic, which is clearly not the case in the patterns just mentioned. In other words, ABC specifically rules these patterns out, because they are long-distance and non-parasitic. The current approach based on feature spreading can account for these patterns, since LA constraints can refer to vowels or root nodes. That is, the distinction between local and long-distance has nothing to do with parasitism.

The third argument refers to some specific properties of ABC. The principal driving force of ABC are two types of markedness constraints: correspondence constraints demand that certain output consonants are in correspondence, whereas the CC identity constraints are satisfied if the said correspondents are identical in terms of a particular feature. These two families of constraints are satisfied equally by spreading/copying the feature or delinking/removing it, as long as all relevant segments are affected. This means that lower ranked faithfulness constraints will always prefer the minimal repair. Imagine an input with three coronals. Both types of markedness constraints will be satisfied by outputs without any retroflexes or with all retroflex coronals. The problem is that the low ranked faithfulness constraint will prefer least changes. This makes the approach susceptible to the 'majority rules'. When most input coronals are retroflex, the output with only retroflex coronals will win. When most input coronals are non-retroflex, the output without retroflexion will win. Such patterns are unattested in assimilation (Baković 2000, Finley 2008). The current approach, on the other hand, cannot generate this pathology.

The final argument comes from trigger-target disparities. In some languages, sets of triggers and targets are disjunctive. Nati in Sankrit (see references above), for example, is triggered by retroflex coronals and targets the coronal nasal. At the same time, coronal nasals cannot be (derived) triggers. This latter phenomenon is known as *icy targets* in the literature (Jurgec 2011a;b, Kimper 2011, Walker 2012). One possible response is to argue that Nati is not a case of consonant harmony (as advocated by Hansson 2001). However, the question whether Nati constitutes consonant harmony (or not) is irrelevant, what matters is that ABC has no way of accounting for trigger-target disparities. There is no doubt that Nati is a case of assimilation, and as such it can also be captured in the current, autosegmental approach (see Jurgec 2011b for further discussion).

We have just seen that ABC can capture only a subset of all assimilation patterns, which means it still

<sup>&</sup>lt;sup>15</sup>In many of these cases, blocking is limited to consonants in a particular prosodic position. This, however, has nothing to do with the inability of ABC to capture blocking in the first place.

requires a spreading account. The current approach, on the other hand, has a broader empirical range and is based on spreading alone.

#### 7. Conclusions

This paper has two main contributions. The empirical contribution is to show that parasitic vowel harmony and consonant harmony are alike. The theoretical contribution is a unified analysis of both phenomena based on feature spreading and two well-established, but somewhat modified constraints—(licensed) alignment and agreement.

The two constraints have very different effects. Alignment constraints prefer spreading to all segments or vowels. This is independently grounded in the general preference of features to be aligned with domain edges and that these features appear in prosodically prominent positions. Agreement constraints, on the other hand, restrict spreading to those targets that are similar to triggers. When the trigger is a vowel, agreement constraints prefer spreading only to those vowels that are similar to the trigger. On the other hand, when the trigger is a consonant, agreement constraints enforce spreading to those consonants that are like the trigger.

The two constraints reveal the common property of all types of parasitic assimilation, which is the additional similarity among the trigger-target pairs. At the same time, the constraints also reveal the difference between vowel and consonant harmony. Since alignment can target vowels, that predicts non-parasitic vowel harmony. In contrast, consonants cannot be targeted by alignment exclusively, hence consonant harmony can only be parasitic. This is in line with the cross-linguistic observations: vowel harmony can be parasitic or not, whereas all reported cases of consonant harmony are parasitic.

#### References

Akinlabi, Akinbiyi. 1996. Featural affixation. Journal of Linguistics 32: 239-289.

- Applegate, Richard Brian. 1972. Inseño Chumash Grammar. Ph.D. thesis, University of California, Berkley, Berkeley, CA.
- Archangeli, Diana. 1984. Underspecification in Yawelmani Phonology and Morphology. Ph.D. thesis, Massachusetts Institute of Technology, Cambridge.

Archangeli, Diana. 1985. Evidence for coplanar representation in nonlinear phonology. Linguistic Inquiry 16 3: 335-372.

Archangeli, Diana. 1988. Underspecification in Yawelmani Phonology and Morphology. Garland, New York.

Archangeli, Diana and Douglas Pulleyblank. 1994. Grounded Phonology. MIT Press, Cambridge, MA.

Archangeli, Diana and Douglas Pulleyblank. 2002. Kinande vowel harmony: domains, grounded conditions and one-sided alignment. *Phonology* 19: 139–188.

Archangeli, Diana and Keiichiro Suzuki. 1997. The Yokuts challenge. In *Derivations and constraints in phonology*, edited by Iggy Roca, pp. 197–226. Oxford University Press, Oxford.

Arsenault, Paul. 2012. Retroflex Consonant Harmony in South Asia. Ph.D. thesis, University of Toronto, Toronto.

Arsenault, Paul and Alexei Kochetov. to appear. Retroflex harmony in Kalasha: Agreement or spreading? In *Proceedings of NELS* 39. GLSA, University of Massachusetts, Amherst. Available on Rutgers Optimality Archive, ROA 1023, http://roa.rutgers.edu.

Baković, Eric. 2000. Harmony, dominance, and control. Ph.D. thesis, Rutgers University, New Brunswick, NJ. Available on Rutgers Optimality Archive, ROA 360, http://roa.rutgers.edu.

Baković, Eric. 2007. Local assimilation and constraint interaction. In *The Cambridge handbook of phonology*, edited by Paul de Lacy, pp. 335–352. Cambridge University Press, Cambridge.

Barnes, Jonathan. 2006. Strength and Weakness at the Interface: Positional neutralization in phonetics and phonology. Mouton de Gruyter, Berlin.

Bashir, Elena. 2003. Dardic. In *The Indo-Iranian Languages*, edited by George Cardona and Dhanesh Jain, pp. 818–894. Routledge, London.

Beckman, Jill N. 1997. Positional faithfulness, positional neutralization and Shona vowel harmony. *Phonology* 14 1: 1–46.

Beckman, Jill N. 1998. *Positional Faithfulness*. Ph.D. thesis, University of Massachusetts, Amherst. Available on Rutgers Optimality Archive, ROA 234, http://roa.rutgers.edu.

Beeler, M. S. 1970. Sibilant harmony in Chumash. International Journal of American Linguistics 36 1: 14–17.

Benua, Laura. 1997. Transderivational Identity: phonological relations between words. Ph.D. thesis, University of Massachusetts, Amherst.

Benus, Stefan. 2005. Dynamics and Transparency in Vowel Harmony. Ph.D. thesis, New York University, New York.

Benus, Stefan and Adamantios I. Gafos. 2007. Articulatory characteristics of Hungarian 'transparent' vowels. *Journal of Phonetics* 35: 271–300.

Bessell, Nicola J. 1998. Local and non-local consonant-vowel interaction in Interior Salish. Phonology 15 1: 1-40.

Blaho, Sylvia. 2008. The syntax of phonology: a radically substance-free approach. Ph.D. thesis, University of Tromsø.

Blumenfeld, Lev A. 2006. *Constraints on phonological interactions*. Ph.D. thesis, Stanford University, Palo Alto, CA. Available on Rutgers Optimality Archive, ROA 877, http://roa.rutgers.edu.

Boyce, Suzanne E. 1990. Coarticulatory organization for lip rounding in Turkish and English. Journal of the Acoustical Society of America 88 6: 2584–2595.

Broselow, Ellen, Su-I Chen, and Marie Huffman. 1997. Syllable weight: convergence of phonology and phonetics. *Phonology* 14: 47–82.

Casali, Roderic F. 1995. Labial opacity and roundness harmony in Nawuri. Natural Language and Linguistic Theory 13: 649-663.

Cho, Young-mee Yu. 1991. On the universality of the coronal articulator. In *Phonetics and phonology 2: volume 2. The special status of coronals, internal and external evidence*, edited by Carole Paradis and Jean-François Prunet, pp. 159–179. Academic Press, San Diego, CA.

Clements, George N. 1976/1980. Vowel harmony in Nonlinear Generative Phonology. Indiana University Linguistics Club, Bloomington.

Clements, George N. 1985a. Akan vowel harmony: a nonlinear analysis. In *African Linguistics: Essays in Honor of W. M. K. Semikenke*, edited by D. L. Goyvaerts, pp. 55–98. John Benjamins, Amsterdam.

Clements, George N. 1985b. The geometry of phonological features. Phonology 2: 225-252.

Clements, George N. 1991. Place of articulation in consonants and vowels: a unified theory. In Working Papers of the Cornell Phonetics Laboratory, vol. 5, pp. 77–123. Cornell University, Ithaca.

Clements, George N. and Engin Sezer. 1982. Vowel and consonant disharmony in Turkish. In *The Structure of Phonological Representations (Part II)*, edited by Harry van der Hulst and Norval Smith, pp. 213–255. Foris, Dordrecht.

Cohn, Abigail C. 1990. *Phonetic and Phonological Rules of Nasalization*. Ph.D. thesis, University of California, Los Angeles, Los Angeles.

Cohn, Abigail C. 1993. The status of nasalized continuants. In *Phonetics and Phonology 2: Nasals, Nasalization, and the Velum*, edited by Marie K. Huffman and Rena A. Krakow, pp. 329–367. Academic Press, San Diego, CA.

Cole, Jennifer. 1987. Planar Phonology and Morphology. Ph.D. thesis, Massachusetts Institute of Technology, Cambridge.

Cole, Jennifer and Charles Kisseberth. 1995. An Optimal Domains Theory of vowel harmony. *Studies in Linguistic Sciences* 24: 101–114. Available on Rutgers Optimality Archive, ROA 22, http://roa.rutgers.edu.

Cole, Jennifer and Loren Trigo. 1988. Parasitic harmony. In *Features, Segmental Structure and Harmony Processes (Part II)*, edited by Harry van der Hulst and Norval Smith, pp. 19–38. Foris, Dordrecht.

Comrie, Bernard. 1981. The Languages of the Soviet Union. Cambridge University Press, Cambridge.

Cook, Eung-Do. 1976. Flattening and rounding in Chilcotin velars. In Victoria Conference on Northwestern Languages, edited by Barbara S. Efrat, pp. 15–32. British Columnia Provincial Museum, Victoria.

Cook, Eung-Do. 1983. Chilcotin flattening. Canadian Journal of Linguistics 28 2: 123-132.

Cook, Eung-Do. 1987. An autosegmental analysis of Chilcotin flattening. In Chicago Linguistic Society (CLS) 23-2: Parasession on Autosegmental and Metrical Phonology, pp. 51–65. Chicago Linguistic Society, Chicago.

Cook, Eung-Do. 1993. Chilcotin flattening and autosegmental phonology. Lingua 91 2/3: 149-174.

Davis, Stuart. 1995. Emphasis spreading in Arabic and Grounded Phonology. Linguistic Inquiry 26 3: 465–498.

Dimmendaal, Gerrit J. 1983. The Turkana Language. Foris, Dordrecht.

Dresher, B. Elan. 2009. The Contrastive Hierarchy in Phonology. Cambridge University Press, Cambridge.

Elfner, Emily. 2011. Syntax-prosody interactions in Irish. Ph.D. thesis, University of Massachusetts, Amherst.

Fanti, Renata. 1938. Note fonetiche e morfologiche sul dialetto di Ascrea (Rieti). L'Italia dialettale 14: 201-218.

Fanti, Renata. 1939. Note fonetiche e morfologiche sul dialetto di Ascrea (Rieti). L'Italia dialettale 15: 101-135.

Fanti, Renata. 1940. Note fonetiche e morfologiche sul dialetto di Ascrea (Rieti). L'Italia dialettale 16: 77-140.

Finley, Sara. 2008. Formal and cognitive restrictions on vowel harmony. Ph.D. thesis, Johns Hopkins University, Baltimore, MD. Available on Rutgers Optimality Archive, ROA 489, http://roa.rutgers.edu.

Gafos, Adamantios I. 1996. *The articulatory basis of locality in phonology*. Ph.D. thesis, Johns Hopkins University, Baltimore, MD. Gafos, Adamantios I. 1996/1999. *The articulatory basis of locality in phonology*. Garland, New York.

Gamble, Geoffrey. 1978. Wikchamni grammar. University of California Press, Berkeley.

Goldsmith, John A. 1976. Autosegmental Phonology. Ph.D. thesis, Massachusetts Institute of Technology, Cambridge.

Goldsmith, John A. 1990. Autosegmental and Metrical Phonology. Blackwell, Oxford.

Goldsmith, John A. 1993. Harmonic phonology. In *The Last Phonological Rule: Reflections on Constraints and Derivations*, edited by John A. Goldsmith, pp. 21–60. University of Chicago Press, Chicago.

Golston, Chris. 1996. Direct Optimality Theory: Representation as pure markedness. Language 72: 713-748.

Gouskova, Maria. 2004. Relational hierarchies in OT: the case of syllable contact. Phonology 21 2: 201-250.

Gouskova, Maria. 2010. The phonology of boundaries and secondary stress in Russian compounds. *The Linguistic Review* 27 4: 387–448.

Hall, Tracy Alan. 1997. The phonology of coronals. John Benjamins, Amsterdam.

Halle, Morris. 1995. Feature geometry and feature spreading. Linguistic Inquiry 16 1: 1-46.

Hamp, Eric P. 1976. Palatalization and harmony in Gagauz and Karaite. In *Tractata altaica: Denis Sinor, sexagenario optime de rebus altaicis merito dedicata*, edited by Walter Heissig, John R. Krueger, Felix J. Oinas, and Edmond Schütz, pp. 211–213. Otto Harrassowitz, Wiesbaden.

Hansson, Gunnar Ólafur. 2001. Theoretical and typological issues in consonant harmony. Ph.D. thesis, University of California, Berkeley.

Hansson, Gunnar Ólafur. 2007a. Blocking effects in Agreement by Correspondence. Linguistic Inquiry 38 2: 395-409.

- Hansson, Gunnar Ólafur. 2007b. On the evolution of consonant harmony: the case of secondary articulation agreement. *Phonology* 24: 77–120.
- Hansson, Gunnar Ólafur. 2010. Long-distance voicing assimilation in Berber: Spreading and/or agreement. In Proceedings of the 2010 Annual Conference of the Canadian Linguistic Association. Concordia University, Montreal.
- Harrington, M. 1974. Sibilants in Ventureño. International Journal of American Linguistics 40 1: 1-9.
- Hayward, Richard J. 1982. Notes on the Koyra language. Afrika und Übersee 65: 211-268.
- Hayward, Richard J. 1988. Remarks on the Omotic sibilants. In *Cushitic-Omotic: Papers from the International Symposium on Cushitic and Omotic Languages*, edited by Marianne Bechhaus-Gerst and Serzisko Fritz, pp. 263–299. Helmut Buske Verlag, Hamburg.
- Hayward, Richard J. 1990. Notes on the Aari language. In *Omotic language studies*, edited by Richard Hayward, pp. 425–493. School of Oriental and African Studies, University of London, London.
- Heegård, Jan and Ida Elisabeth Mørch. 2004. Retroflex vowels and other peculiarities in the Kalasha sound system. In *Himalayan languages: past and present*, edited by Anju Saxena, pp. 57–76. Mouton de Gruyter, Berlin.

Herbert, Raymond J. and Nicholas N. Poppe. 1963. Kirghiz Manual. Indiana University, Bloomington, The Hague.

- Howard, Irwin. 1972. A directional theory of rule application in phonology. Ph.D. thesis, Massachusetts Institute of Technology, Cambridge.
- Hualde, Jose Ignacio. 1998. Asturian and Cantabrian metaphony. Rivista di Linguistica 10: 99-108.
- van der Hulst, Harry and Norval Smith. 1982. Prosodic domains and opaque segments in Autosegmental Theory. In *The Structure of Phonological Representations (Part II)*, edited by Harry van der Hulst and Norval Smith, pp. 311–336. Foris, Dordrecht.
- Hyde, Brett. 2012. Alignment constraints. Natural Language and Linguistic Theory 30 3: 789-836.
- Itô, Junko. 1984. Melodic dissimilation in Ainu. Linguistic Inquiry 15: 505–513.
- Itô, Junko. 1986/1988. Syllable theory in prosodic phonology. Garland, New York.
- Itô, Junko and Armin Mester. 1995. The core-periphery structure of the lexicon and constraints on reranking. In University of Massachusetts Occasional Papers in Linguistics 18, edited by Jill Beckman, Suzanne Urbanczyk, and Laura Walsh, pp. 181–209. GLSA, University of Massachusetts, Amherst.
- Itô, Junko, Armin Mester, and Jaye Padgett. 1995. Licensing and underspecification in Optimality Theory. *Linguistic Inquiry* 26 4: 571–631.

Jensen, John T. 1974. A constraint on variables in phonology. Language 50: 675-686.

Jensen, John T. and Margaret Stong-Jensen. 1979. The Relevancy Condition and variables in phonology. *Linguistic Analysis* 5 2: 125–160.

Johnson, Douglas C. 1972. Formal aspects of phonological description. Mouton de Gruyter, The Hague and Paris.

- Johnson, Douglas C. 1980. Regular disharmony in Kirgiz. In *Issues in vowel harmony*, edited by Robert Vago, pp. 89–99. John Benjamins, Amsterdam.
- Jurgee, Peter. 2011a. Feature Spreading 2.0: A Unified Theory of Assimilation. Ph.D. thesis, University of Tromsø, Tromsø. Available on LingBuzz, http://ling.auf.net/lingBuzz/001281.
- Jurgec, Peter. 2011b. Icy targets. Ms. Meertens Institute, Amsterdam.
- Jurgec, Peter. 2012a. Long-distance derived environment effects. Ms. Leiden University.
- Jurgec, Peter. 2012b. Walker (2011). Vowel patterns in language. (Cambridge Studies in Linguistics 130.) Cambridge: Cambridge University Press. Pp. x + 356. *Phonology* 29 3: 533–539.
- Kaplan, Aaron. 2008. Noniterativity is an emergent property of grammar. Ph.D. thesis, University of California, Santa Cruz. Available on Rutgers Optimality Archive, ROA 977, http://roa.rutgers.edu.
- Kaun, Abigail R. 1995. The typology of rounding harmony: an Optimality Theoretic approach. Ph.D. thesis, University of California, Los Angeles. Available on Rutgers Optimality Archive, ROA 227, http://roa.rutgers.edu.
- Kaun, Abigail R. 2004. The typology of rounding harmony. In *Phonetically based phonology*, edited by Bruce Hayes, Robert Kirchner, and Donca Steriade, pp. 87–116. Cambridge University Press, Cambridge.

Kenstowicz, Michael. 1997. Quality-sensitive stress. Rivista di Linguistica 9 1: 157-187.

- Kimper, Wendell. 2011. Competing Triggers: Transparency and Opacity in Vowel Harmony. Ph.D. thesis, University of Massachusetts, Amherst.
- Kiparsky, Paul. 1981. Vowel harmony. Ms. Massachusetts Institute of Technology.

Kiparsky, Paul. 1985. Some consequences of Lexical Phonology. Phonology 2 3: 85-138.

- Kirchner, Robert. 1993. Turkish vowel harmony and disharmony: An Optimality Theoretic account. Ms. University of Alberta. Available on Rutgers Optimality Archive, ROA 4, http://roa.rutgers.edu.
- Kisseberth, Charles. 1969. Theoretical implications of Yawelmani phonology. Ph.D. thesis, University of Illinois, Urbana, IL.
- Kowalski, Tadeusz. 1929. Karaimische texte im dialekt von Troki. Nakładem Polskiej Akademji Umiejętności, Cracow.
- Krämer, Martin. 2001. Yucatec Maya vowel alternations harmony as syntagmatic identity. Zeitschrift für Sprachwissenschaft 20: 175–217.

Krämer, Martin. 2003. Vowel harmony and Correspondence Theory. Mouton de Gruyter, Berlin.

- Kuroda, S.-Y. 1967. Yawelmani Phonology. MIT Press, Cambridge, MA.
- de Lacy, Paul. 2001. Markedness in prominent positions. In *MIT Working Papers in Linguistics 40: HUMIT 2000*, edited by Ora Matushansky, Albert Costa, Javier Martin-Gonzales, Lance Nathan, and Adam Szczegielniak, pp. 53–66. MITWPL, Cambridge, MA. Available on Rutgers Optimality Archive, ROA 432, http://roa.rutgers.edu.
- de Lacy, Paul. 2002. The interaction of tone and stress in Optimality Theory. Phonology 19 1: 1-32.

de Lacy, Paul. 2004. Markedness conflation in Optimality Theory. Phonology 21 1: 1-55.

de Lacy, Paul. 2006. Markedness: reduction and preservation in phonology. Cambridge University Press, Cambridge.

de Lacy, Paul. 2007. The interaction of tone, sonority, and prosodic structure. In *The Cambridge handbook of phonology*, edited by Paul de Lacy, pp. 281–307. Cambridge University Press, Cambridge.

Ladefoged, Peter and Ian Maddieson. 1996. The Sounds of the World's Languages. Blackwell, Malden, MA.

Langendoen, Terrence D. 1968. The London School of Linguistics. MIT Press, Cambridge, MA.

Lieber, Rochelle. 1987. An Integrated Theory of Autosegmental Processes. State University of New York Press, Albany, NY.

Lombardi, Linda. 1998. Evidence for Max Feature constraints from Japanese. In University of Maryland Working Papers in Linguistics 7, edited by Haruka Fukazawa, Frida Morelli, Caro Struijke, and Yi-Ching Su, pp. 41–62. University of Maryland. Available on Rutgers Optimality Archive, ROA 247, http://roa.rutgers.edu.

Lombardi, Linda. 1999. Positional Faithfulness and voicing assimilation in Optimality Theory. Natural Language and Linguistic Theory 17 2: 267–302.

Mackenzie, Sara. 2009. Contrast and Similarity in Consonant Harmony Processes. Ph.D. thesis, University of Toronto, Toronto.

Mahanta, Shakuntala. 2008. Directionality and locality in vowel harmony: with special reference to vowel harmony in Assamese. Landelijke Onderzoekschool Taalwetenschap (LOT), Utrecht.

Maiden, Martin. 1991. Interactive Morphology: Metaphony in Italian. Routledge, New York.

McCarthy, John J. 1997. Process specific constraints in Optimality Theory. Linguistic Inquiry 28: 231-251.

McCarthy, John J. 1999. Sympathy and phonological opacity. *Phonology* 16 3: 331–399.

McCarthy, John J. 2003. OT constraints are categorical. Phonology 20: 75-138.

- McCarthy, John J. 2007a. Consonant harmony via correspondence: Evidence from Chumash. In *University of Massachusetts Occasional Papers in Linguistics 32: Papers in Optimality Theory III*, edited by Leah Bateman, Michael O'Keefe, Ehren Reilly, and Adam Werle, pp. 223–238. GLSA, University of Massachusetts, Amherst. Available on Rutgers Optimality Archive, ROA 879, http://roa.rutgers.edu.
- McCarthy, John J. 2007b. *Hidden Generalizations: Phonological Opacity in Optimality Theory*. Equinox Publishing Company, London.

McCarthy, John J. 2009. Harmony in Harmonic Serialism. Ms., University of Massachusetts, Amherst. Available on Rutgers Optimality Archive, ROA 1009, http://roa.rutgers.edu.

- McCarthy, John J. and Alan Prince. 1993. Generalized Alignment. In Yearbook in Morphology 1993, edited by Geert Booij and Jaap van Marle, pp. 79–153. Kluwer, Dordrecht.
- Mørch, Ida Elisabeth and Jan Heegård. 1997. Retroflekse vokalers oprindelse i kalashamon i historisk og areallingvistisk perspektiv og Variation i sprogbeskrivelsen: vokallængde i kalashamon: En analyse af to emner i kalashamon-fonologien. Master's thesis, University of Kopenhagen, Kopenhagen.
- Morén, Bruce. 1999/2001. Distinctiveness, coercion and sonority: a unified theory of weight. Routledge, New York. Available on Rutgers Optimality Archive, ROA 349, http://roa.rutgers.edu.

Morén, Bruce. 2000. The puzzle of Kashmiri stress: implications for weight theory. Phonology 17: 365-396.

Morén, Bruce. 2003. The parallel structures model of feature geometry. In *Working Papers of the Cornell Phonetics Laboratory*, edited by J. Brugman and A. Rhiel, vol. 15, pp. 194–270. Cornell University.

Morgenstierne, Georg. 1973. Indo-Iranian frontier languages: Volume 4: The Kalasha Language. Universitetsforlaget, Oslo.

Mullin, Kevin. 2011. Strength in harmony systems: trigger and directional asymmetries. Ms. University of Massachussets, Amherst. Available on Rutgers Optimality Archive, http://roa.rutgers.edu, ROA 1134.

Myers, Scott. 1997. OCP effects in Optimality Theory. Natural Language and Linguistic Theory 15 4: 847-892.

Myrberg, Sara. 2010. The Intonational Phonology of Stockholm Swedish. Ph.D. thesis, University of Stockholm.

Nevins, Andrew. 2010. Locality in Vowel Harmony. MIT Press, Cambridge, MA.

Nevins, Andrew and Bert Vaux. 2004. Consonant harmony in Karaim. In MIT Working Papers in Linguistics 46: The Proceedings of the Workshop on Altaic in Formal Linguistics, edited by Anikó Csirmaz, Youngjoo Lee, and MaryAnn Walker, pp. 175–194. MITWPL, Cambridge, MA.

Newman, Stanley S. 1944. The Yokuts Language of California. The Viking Fund Publications in Anthropology, New York.

- Ní Chiosáin, Máire and Jaye Padgett. 1997. Markedness, segment realization, and locality of spreading. In *Report no. LRC-97-01, Linguistics Research Center*. University of California, Santa Cruz. Available on Rutgers Optimality Archive, ROA 188, http://roa.rutgers.edu.
- Ní Chiosáin, Máire and Jaye Padgett. 2001. Markedness, segment realization, and locality of spreading. In Segmental phonology in Optimality Theory: constraints and representations, edited by Linda Lombardi, pp. 118–156. Cambridge University Press, Santa Cruz.

Noske, Manuela. 1990. Vowel harmony in Turkana. Studies in the Linguistic Sciences 20: 123-134.

Noske, Manuela. 1996. [ATR] harmony in Turkana. Studies in African Linguistics 25 1: 61-99.

Noske, Manuela. 2000. [ATR] harmony in Turkana: A case of Faith suffix  $\gg$  Faith root. *Natural Language and Linguistic Theory* 18: 771–812.

Noske, Roland. 1985. Syllabification and syllable changing processes in Yawelmani. In Advances in Nonlinear Phonology, edited by Harry van der Hulst and Norval Smith, pp. 335–361. Foris, Dordrecht.

Noske, Roland. 1993. A Theory of Syllabification and Segmental Alternation: With studies on the phonology of French, German, Tonkawa and Yawelmani. Niemeyer, Tübingen.

Odden, David. 1994. Adjacency paramaters in phonology. Language 70 2: 289-330.

Orgun, Orhan and Ronald Sprouse. 2007. A freer input: Yowlumne opacity and the Enriched Input Model. In Freedom of Analysis?,

edited by Sylvia Blaho, Patrik Bye, and Martin Krämer, pp. 93-121. Mouton de Gruyter, Berlin.

Paster, Mary. 2004. Vowel height harmony and blocking in Buchan Scots. Phonology 21: 359-407.

- Pater, Joe. 1999. Austronesian nasal substitution and other NC effects. In *The prosody—morphology Interface*, edited by René Kager, Harry van der Hulst, and Wim Zonneveld, pp. 310–343. Cambridge University Press, Cambridge, MA. Available on Rutgers Optimality Archive, ROA 160, http://roa.rutgers.edu.
- Piggott, Glyne. 1992. Variability in feature dependency: The case of nasality. Natural Language and Linguistic Theory 10: 33-77.

Piggott, Glyne. 2003. Theoretical implications of segment neutrality in nasal harmony. *Phonology* 20 3: 375–424.

Piggott, Glyne and Harry van der Hulst. 1997. Locality and the nature of nasal harmony. Lingua 103: 85-112.

- Poser, William J. 1982. Phonological representation and action-at-a-distance. In *The structure of phonological representations (part II)*, edited by Harry van der Hulst and Norval Smith, pp. 121–158. Foris, Dordrecht.
- Poser, William J. 1993. Are strict cycle effects derivable? In *Studies in Lexical Phonology*, edited by Sharon Hargus and Ellen M. Kaisse, pp. 315–321. Academic Press, New York.

Poser, William J. 2004. On the status of Chumash sibilant harmony. Ms., University of Pennsylvania, Philadelphia.

- Prince, Alan and Paul Smolensky. 1993/2004. Optimality Theory: constraint linteraction in Generative Grammar. Blackwell, Malden, MA. Available on Rutgers Optimality Archive, ROA 537, http://roa.rutgers.edu.
- Pulleyblank, Douglas. 1996. Neutral vowels in Optimality Theory: a comparison of Yoruba and Wolof. *Canadian Journal of Linguistics* 41 1: 295–347.

Raimy, Eric. 2000. The phonology and morphology of reduplication. Mouton de Gruyter, Berlin.

- Rhodes, Russell. 2011. Vowel harmony as Agreement by Correspondence. Ms. University of California Berkeley. Available online at http://linguistics.berkeley.edu/~russellrhodes/pdfs/abc\_vh.pdf [Aug-3-12].
- Ringen, Catherine and Robert Vago. 1998. Hungarian vowel harmony in Optimality Theory. Phonology 15 3: 393-416.
- Robins, R. H. 1957. Vowel nasality in Sundanese, a phonological and grammatical study. In *Studies in linguistic analysis*, pp. 87–103. Blackwell, Oxford.
- Rose, Sharon. 2004. Long-distance vowel-consonant agreement in Harari. *Journal of African Languages and Linguistics* 25 1: 41–87. Rose, Sharon and Rachel Walker. 2004. A typology of consonant agreement as correspondence. *Language* 80 3: 475–531.
- Sagey, Elizabeth. 1990. The representation of features in non-linear phonology: the articulator node hierarchy. Garland, New York. Samek-Lodovici, Vieri and Alan Prince. 1999. Optima. Rutgers Center for Cognitive Science, RuCCS-TR-57. Available on Rutgers
- Optimality Archive, ROA 363, http://roa.rutgers.edu.
- Samek-Lodovici, Vieri and Alan Prince. 2002. Fundamental properties of harmonic bounding. Rutgers Center for Cognitive Science, RuCCS-TR-71. Available on Rutgers Optimality Archive, ROA 785, http://roa.rutgers.edu.
- Schein, Barry and Donca Steriade. 1986. On geminates. Linguistic Inquiry 17 4: 691-744.
- Selkirk, Elisabeth. 1980. Prosodic domains in phonology: Sanskrit revisited. In Juncture: a collection of original papers, edited by Mark Aronoff and Mary-Louise Kean, pp. 107–129. Anma Libri, Saratoga, CA.
- Selkirk, Elisabeth. 2011. The syntax-phonology interface. In *The Handbook of Phonological Theory*, edited by John A. Goldsmith, Jason Riggle, and Alan C. L. Yu, pp. 435–484. Blackwell, Malden, MA, 2nd edn.
- Shaw, Patricia A. 1991. Consonant harmony systems: The special status of coronal harmony. In *The special status of coronals:* Internal and external evidence, edited by Carole Paradis and Jean-François Prunet, pp. 125–158. Academic Press, San Diego, CA.
- Smolensky, Paul. 1993. Harmony, markedness, and phonological activity. Paper presented at Rutgers Optimality Workshop 1, Rutgers University. Available on Rutgers Optimality Archive, ROA 87, http://roa.rutgers.edu.
- Sprouse, Ronald. 1997. A case for Enriched Inputs. Ms., University of California Berkeley. Available on Rutgers Optimality Archive, ROA 193, http://roa.rutgers.edu.
- Steriade, Donca. 1981. Parameters of metrical vowel harmony rules. Ms., Massachusetts Institute of Technology.
- Steriade, Donca. 1986. Yokuts and the vowel place. Linguistic Inquiry 17 1: 129–146.
- Steriade, Donca. 1995. Underspecification and markedness. In *The handbook of phonological theory*, edited by John A. Goldsmith, pp. 114–174. Blackwell, Oxford.
- Steriade, Donca. 2001. Directional assymetries in place assimilation: a perceptual account. In *The role of speech perception in phonology*, edited by Elizabeth V. Hume and Keith Johnson, pp. 219–250. Academic Press, New York.
- Steriade, Donca. 2001/2008. The phonology of perceptibility effects: the P-map and its consequences for constraint organization. In The nature of the word: essays in honor of Paul Kiparsky, edited by Kristin Hanson and Sharon Inkelas, pp. 151–180. MIT Press, Cambridge, MA.

Trail, Robert L. n.d. Kalasha-English Dictionary. NIPS-SIL, Islamabad.

- Uffmann, Christian. 2006. Epenthetic vowel quality in loanwords: Empirical and formal issues. Lingua 116: 1079–1111.
- Vago, Robert and Harry Leder. 1987. On the autosegmental analysis of vowel harmony in Turkana. In Current Approaches to African Linguistics: Volume 4, edited by David Odden, pp. 383–395. Foris, Dordrecht.
- Vaysman, Olga. 2009. Segmental Alternations in Metrical Theory. Ph.D. thesis, Massachusetts Institute of Technology, Cambridge. Available on Rutgers Optimality Archive, ROA 1011, http://roa.rutgers.edu.

Walker, Rachel. 1998/2000. Nasalization, neutral segments, and opacity effects. Garland, New York.

- Walker, Rachel. 2000. Yaka nasal harmony: Spreading or segmental correspondence? In *Proceedings of BLS 26*, edited by Lisa J. Conathan, Jeff Good, Darya Kavitskaya, Alyssa B. Wulf, and Alan C. L. Yu, pp. 321–332. Berkeley Linguistics Society, Berkeley, CA.
- Walker, Rachel. 2001. Round licensing, harmony, and bisyllabic triggers in Altaic. *Natural Language and Linguistic Theory* 19: 827–878.

Walker, Rachel. 2004. Vowel feature licensing at a distance: Evidence from northern Spanish language varieties. In Proceedings of the

23<sup>rd</sup> West Coast Conference on Formal Linguistics, edited by Benjamin Schmeister, Vineeta Chand, Ann Kelleher, and Angelo J. Rodríguez, pp. 787–800. Cascadilla Press, Somerville, MA.

Walker, Rachel. 2005. Weak triggers in vowel harmony. Natural Language and Linguistic Theory 23 4: 917-989.

Walker, Rachel. 2011. Vowel patterns in language. Cambridge University Press, Cambridge.

Walker, Rachel. 2012. Vowel harmony in Optimality Theory. Language and Linguistics Compass 6: 575-592.

Walker, Rachel, Dani Byrd, and Fidèle Mpiranya. 2008. An articulatory view of Kinyarwanda coronal harmony. *Phonology* 25: 499-535.

Walker, Rachel and Geoffrey Pullum. 1999. Possible and impossible segments. Language 75: 764-780.

Wayment, Adam. 2009. Assimilation as Attraction: Computing Distance, Similarity, and Locality in Phonology. Ph.D. thesis, Johns Hopkins University, Baltimore, MD.

Wilson, Colin. 2000. Targeted constraints: an approach to contextual neutralization in Optimality Theory. Ph.D. thesis, Johns Hopkins University, Baltimore, MD.

Wilson, Colin. 2001. Consonant cluster neutralization and targeted constraints. *Phonology* 18 1: 147–197.

Wurm, Stefan. 1949. The (Kare-)Kirghiz language. Bulletin of the School of Oriental and African Studies, University of London 13 1: 97–120.

Zec, Draga. 1988/1994. Sonority constraints on prosodic structure. Garland, New York.

Zec, Draga. 1995. Sonority constraints on prosodic structure. *Phonology* 12: 85–129.

Zoll, Cheryl. 1993. Directionless syllabification and ghosts in Yawelmani. Ms., University of California Berkeley. Available on Rutgers Optimality Archive, ROA 143, http://roa.rutgers.edu.

Zoll, Cheryl. 1998a. Parsing below the segment in a constraint-based framework. CSLI Publications, Stanford, CA.

Zoll, Cheryl. 1998b. Positional asymmetries and licensing. Ms., Massachusetts Institute of Technology. Available on Rutgers Optimality Archive, ROA 282, http://roa.rutgers.edu.